Author’s response to reviews

Title: Enthusiasm for homework and improvement in subthreshold depression during behavior therapy: Secondary analysis of data from a randomized controlled trial

Authors:

Yu Hayasaka (y.hy.hh.y3@gmail.com)
Toshi A Furukawa (furukawa@kuhp.kyoto-u.ac.jp)
Takashi Sozu (sozu.takashi.4s@kyoto-u.ac.jp)
Hissei Imai (ihits@hotmail.com)
Norito Kawakami (norito@m.u-tokyo.ac.jp)
Masaru Horikoshi (mhorikoshi335@gmail.com)

Version: 4 Date: 17 November 2015

Author’s response to reviews: see over
2015/11/17
Dr. Anna Clark
Editor-in-Chief
BMC Psychiatry

Ms. Ref. No: 1077761307119974

Dear Dr. Clark,

Thank you very much for your reviews of our manuscript referenced above and for allowing us to resubmit it for publication. We have now revised our manuscript in accordance with your reviewers’ comments. Our responses are presented in details on the following pages. We also attached the revised manuscript with the main changes shown in red letters following our responses to the reviewers’ comments.

We hope that the revised version of our paper is now acceptable for publication.

Sincerely yours,

Yu Hayasaka
Department of Health Promotion and Human Behavior, Kyoto University Graduate School of Medicine / School of Public Health, Yoshida Konoe-cho, Sakyo-ku, Kyoto 606-8501, Japan
Phone: +81-75-753-9491
Fax: +81-75-753-4641
E-mail: y.hy.hh.y3@gmail.com
Editorial requirements:

1) Please include details of the consent procedures for the original RCT.

Reply: We have added information of consent procedure in Material and Method as follows; “At the initial screening, potential participants were asked to provide their written informed consent to fill in the screening questionnaire after full explanation of the purpose and procedure of the study. Then those who fulfilled the eligibility criteria were asked to provide their final informed consent to participate in the intervention study.”

2) Trial Registration Number and date at the end of the abstract.

Reply: We have added Trial Registration Number and date at the end of the abstract.

3) CONSORT guidelines

In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/about/editorialpolicies#StandardsofReporting), could you please ensure your manuscript reporting adheres to CONSORT guidelines (http://www.consort-statement.org/) for reporting clinical trials. This is so your methodology can be fully evaluated and utilised. Can you please include a completed CONSORT checklist as an additional file when submitting your revised manuscript. We would also ask that you include a completed copy of the CONSORT flowchart for your study as a figure in your manuscript.

Reply: As I contacted by email, it is difficult to include most information of CONSORT checklist in our manuscript. According to your explanation in the email, we have added the sentence in Material and Method; “The original report of this RCT adhered to CONSORT guidelines”.

4) Please confirm that all provide information in Competing Interest represents a true Competing Interest for this study.

Reply: I have confirmed the information of Competing Interest and added a sentence; “HY, IH and HM have no competing interests to declare.”
Enthusiasm for homework and improvement of psychological distress in subthreshold depression during behavior therapy: Secondary analysis of data from a randomized controlled trial

Hayasaka Y1§, Furukawa TA1, Sozu T2, Imai H1, Kawakami N3, Horikoshi M4,
On behalf of the GENKI Project"

1 Department of Health Promotion and Human Behavior, Kyoto University Graduate School of Medicine / School of Public Health, Yoshida Konoe-cho, Sakyo-ku, Kyoto, 606-8501, Japan
2 Department of Management Science, Faculty of Engineering, Tokyo University of Science, 1-3 Kagurazaka, Shinjyuku-ku, Tokyo, 162-8601, Japan
3 Department of Mental Health, University of Tokyo Graduate School of Medicine, Hongo, Bunkyo-ku, Tokyo, 113-0033, Japan
4 National Center of Neurology and Psychiatry, Kodaira, Tokyo, 187-8551, Japan

§Corresponding author
" Membership of the GENKI Project is provided in the Acknowledgments.

Email addresses:

HY: hayasaka.yu.48m@st.kyoto-u.ac.jp
TAF: furukawa@kuhp.kyoto-u.ac.jp
ST: sozu@ms.kagu.tus.ac.jp
IH: ihits@hotmail.com
KN: norito@m.u-tokyo.ac.jp
HM: mhorikoshi335@gmail.com
Background:
Cognitive behavioral therapy (CBT) usually involves homework, the completion of which is a known predictor of a positive outcome. The aim of the present study was to examine the session-by-session relationships between enthusiasm to complete the homework and the improvement of psychological distress in depressed people through the course of therapy.

Methods:
Working people with subthreshold depression were recruited to participate in the telephone CBT (tCBT) program with demonstrated effectiveness. Their enthusiasm for homework was enhanced with motivational interviewing techniques and was measured by asking two questions: “How strongly do you feel you want to do this homework?” and “How confident do you feel you can actually accomplish this homework?” at the end of each session. The outcome was the K6 score, which was administered at the start of each session. The K6 is an index of psychological distress including depression and anxiety. We used structural equation modeling (SEM) to elucidate the relationships between enthusiasm and the K6 scores from session to session.

Results:
The best fitting model suggested that, throughout the course of behavior therapy (BT), enthusiasm to complete the homework was negatively correlated with the K6 scores for the subsequent session, while the K6 score measured at the beginning of the session did not influence the enthusiasm to complete the homeworks assigned for that session.

Conclusions:
Empirical data now support the practitioners of BT when they try to enhance their patient’s enthusiasm for homework regardless of the participant’s distress, which then would lead to a reduction in distress in the subsequent week.

**Trial Registration Number and date of original RCT:**

**Keywords:**

Enthusiasm, Cognitive behavioral therapy, Telephone cognitive behavioral therapy, Homework, Depression, Subthreshold depression, Structural equation modeling,
BACKGROUND

Subthreshold depression, sometimes also called subsyndromal depression or minor depression, refers to a depressive state that does not meet the full diagnostic criteria for major depressive disorder (MDD) [1]. It would be classified as Depressive Disorder Not Otherwise Specified in DSM-IV [2] or Other Specified Depressive Disorder in DSM-5 [3]. Patients with subthreshold depression often present with both depression and anxiety symptoms, albeit both under any diagnostic thresholds [4]. It has been increasingly recognized that subthreshold depression is highly prevalent in the general population [5] and in the primary care [6], is clinically relevant because it significantly affects the quality of life and functioning of the sufferers [7] and carries a high risk of developing major depressive disorder [8], and is societally important because it is associated with high economic costs [9].

Cognitive behavioral therapy (CBT), the psychotherapy for depression and anxiety with the largest number of randomized controlled trials (RCTs) supporting its efficacy [10, 11], has been shown to be effective for subthreshold depression [12] as well. In CBT, patients self-monitor their own behaviors, emotions and thoughts and also practice newly learned cognitive or behavioral coping skills outside the therapy sessions as homework. Homework is usually assigned in every session of CBT, asking patients to practice newly learned cognitive or behavioral skills and to generalise such skills to their daily situations in which their problems occur [13]. In general, homework has been shown to facilitate improvement in depression, anxiety or other client problems through CBT [14] but patients often find it difficult to complete the homework [15-19].

In order to understand and possibly facilitate the process of CBT, it is therefore important to examine which aspects of the homework are important in this therapeutic process. Client characteristics, therapist characteristics, characteristics of the task, and interrelationships among these components have been considered to influence the homework compliance [20]. Among the first components, the clients’ motivational level
may be one of the important elements for enhancing homework compliance. Some psychological interventions, such as motivational interviews, are aimed at enhancing enthusiasm [21]. However, no study has yet examined whether the enthusiasm to complete homework assignments thus enhanced can lead to improved process and outcome of CBT.

Moreover, the extant studies of homework in CBT suffers one crucial methodological weakness. In the literature, homework compliance has usually been assessed either post-hoc after the treatment is over, thus risking the recall bias [22, 23], or only once out of the 10 or more sessions of the program, thus possibly not reflecting the overall compliance [24, 25]. Even when homework compliance was measured several times, only the average of those several values was used to predict the outcome of the treatment [26, 27]. To the best of our knowledge, no study to date has examined the session-by-session relationships between homework and an improvement through CBT.

We have previously conducted a randomized controlled trial of CBT administered via telephone among employees with subthreshold depression in a large company in Japan. This telephone CBT (tCBT) program was shown to have a large effect, in comparison with a waiting list control group, with an effect size of around 0.7 for the primary outcome of general psychological distress including depression and anxiety [28]. In this trial, we applied motivational interviewing techniques and measured the enthusiasm of the participants for completing homework at the end of each session.

The current study therefore focused on enthusiasm rather than compliance and aimed to examine the session-by-session relationships between the participants’ enthusiasm to complete the homework and the therapeutic outcomes using structural equation modeling (SEM). We hypothesized that high enthusiasm to complete the homework assignments would be associated with improvement of psychological distress in the subsequent session.
MATERIALS AND METHODS

This study is a secondary analysis of data from a previous RCT. The original RCT was approved by the Ethics Review Committee of Nagoya City University Graduate School of Medicine, and the present re-analysis has been approved by the Ethics Committee of Kyoto University Graduate School of Medicine. The original report of this RCT adhered to CONSORT guidelines and the research design and methods of the RCT have been described in detail in a previous publication [28]. Summarized briefly, the study involved an RCT to compare the effectiveness of tCBT in addition to an Employee Assistance Program (EAP) and EAP alone for the treatment of subthreshold depression among workers at a large manufacturing company in Japan. At the initial screening, potential participants were asked to provide their written informed consent to fill in the screening questionnaire after full explanation of the purpose and procedure of the study. Then those who fulfilled the eligibility criteria were asked to provide their final informed consent to participate in the intervention study. Two thousand three hundred twenty-two employees were assessed for eligibility, of whom 145 met the eligibility criteria and invited to participate in the RCT. Twenty-seven declined and 118 finally took part in the original RCT. Fifty-eight were allocated to the tCBT arm and started the telephone sessions immediately. Sixty were allocated to the EAP alone arm (waiting list) and had to wait for four months, complete the end-of-treatment questionnaire and, if they still desired, began receiving the tCBT sessions. (See Appendix Figure 1)

Participants

We defined subthreshold depression as depressive state scoring above the predefined thresholds on screening questionnaires but failing to reach the diagnostic threshold for major depression.

The inclusion criteria were as follows:
1) An age of 20–57 years at the time of study entry, (the retirement age was 60 and we took a period for follow-up into consideration)

2) Currently employed full-time (either regular or temporary),

3) Expected to be employed full-time for 6 months after the screening,

4) K6 scores greater than or equal to 9 at screening and greater than or equal to 5 at the first tCBT session (baseline),

5) Beck Depression Inventory-II (BDI2) [29] score of greater than or equal to 10 at screening, and

6) Participation in one or more sessions of the tCBT program.

The exclusion criteria were as follows:

1) Major depressive episode in the past month, as ascertained using the Composite International Diagnostic Instrument (CIDI) [30] (We did not exclude dysthymia or major depression in partial remission.),

2) Lifetime history of bipolar disorder, as ascertained using the CIDI,

3) Any substance dependence during the past 12 months, as ascertained using the CIDI,

4) Any other current mental disorder if it constituted the predominant aspect of the clinical presentation and required treatment not offered in the study,

5) Current treatment for a mental health problem from a mental health professional,

6) Sick leave for 6 or more days for a physical or mental condition in the past month, and

7) Expected to be on pregnancy leave, maternity leave, or nursing leave within 6 months after screening.

**Measures**

**Enthusiasm**

The participants’ enthusiasm to complete therapy-related homework was estimated from the responses to two questions. The therapist asked the participants at the end of
each session after setting the homework for the week, “How strongly do you feel you want to do this homework? (Q1)” and “How confident do you feel you can actually accomplish this homework this week?” (Q2). The participants responded using a scale of 0 to 10, with 0 representing no enthusiasm and 10 representing maximum enthusiasm. When there were more than one homework assignment from the session, the two questions were repeated for each homework assignment and the average score was used to represent the participants’ enthusiasm at that session.

In asking these questions, some motivational interviewing techniques were used, so that difficult homework assignments were broken down to smaller pieces and/or tips to troubleshoot potential barriers to completing homework were discussed between the therapist and the participant using Socratic questions (e.g., When the patient rating was high, for example, at 8, a Socratic question such as “I can see that you are quite willing/confident to do this homework. But what is in the remaining 2 points?” might further enhance the motivation of the participant and/or would call for collaborative troubleshooting for possible barriers to homework completion. When the rating was moderate, for example, at 5, a different question such as “What can we do to improve your rating by one or two points?” can be very helpful for both the participant and the therapist to collaboratively modify and clarify the homework task. And when the rating was very low, for example 2 or even 1, then a complete breaking down of the homework task would be called for.). The final scores were collected after the participants’ enthusiasm for the week’s homework would be enhanced as much as possible through motivational interviewing.

Cronbach’s alpha coefficient for the enthusiasm measure was 0.75 (95%CI: 0.68 to 0.80). In general alpha coefficients of 0.7 or greater are considered to be represent satisfactory internal consistency reliability [31].

Psychological distress
The outcome was the K6 score, measured at the beginning of each session regarding the participant’s status for the preceding week. The K6 is a 6-item, self-reported index of psychological distress, including four questions about depression (hopelessness, depressed mood, anergia, and worthlessness) and two questions about anxiety (nervousness and restlessness) [32]. Each item is rated between 0=none of the time and 4=all of the time. The total score therefore ranged between 0 and 24. The Japanese version has been validated [33]. We administered the K6 as an initial screening instrument and used a cutoff of 8/9 according to the Japanese calibration study [33]. We also used K6 as a process measure throughout the course of tCBT. K6 usually measures psychological distress for the past 30 days but we modified this time frame to the past seven days in this RCT. K6 thus measured showed a Pearson correlations coefficient of 0.62 (95%CI: 0.53 to 0.69, p<0.001) with the standard self-rating depression instrument BDI-II and demonstrated similar sensitivity to change in this study [28].

Treatment

The telephone CBT was a structured, manualised, eight-session program adapted from a previously established manual [34]. The participant and the therapist shared the patient manual containing all the materials for each session. A separate therapist manual was prepared specifying the procedure for each session. The participant and the therapist also shared a handbook that contained all the homework worksheets in a small notebook format.

Each session was designed to last 30 to 45 minutes, but the actual lengths varied according to the participants’ needs and the therapists’ assessment. Sessions were carried out at weekly intervals, but the scheduling was flexible. Each session began with an assessment of the participant’s distress symptoms during the past week using the K6 and a review of the previous session and the homework. The first session included psychoeducation regarding the CBT model and explained the theory of the whole
program and gave the homework to self-monitor one’s own mood and associated thoughts. Sessions #2 through #4 focused on increasing pleasant activities through behavioral therapy (BT) [35]. Sessions #5 through #7 focused on cognitive therapy (CT) of negative automatic thoughts [36]. In Session #8 the participant and the therapist reviewed the behavioral and cognitive skills covered in the program and created a personal self-care plan. All the sessions included an assessment of enthusiasm in completing homework assignments in their daily lives, i.e. each participant was asked the two questions explained above for each homework.

The telephone therapists were clinical psychologists, social workers or nurses with at least 1 year of clinical experience. They were required to receive a didactic lecture, to listen to audiotaped sessions, and to treat two clients under supervision before they could participate as therapists. In addition, the quality of the administered tCBT sessions was assured by on-going supervision and consultation. Each participant received all the sessions from the same therapist. Because the original RCT compared the addition of tCBT to usual care against usual care alone, participants in both conditions were allowed to utilise the EAP provided by the company, which included stress diagnostics and a reduction program on the Internet, telephone consultation, and email consultation, when the employee desired. They were also free to seek any professional help, such as professional counselors and medical doctors. During tCBT, however, only one participant used telephone consultation once.

Data analysis

We first summarised demographic and clinical characteristics of the participants at baseline and investigated if these variables (i.e., gender, age, job category, and job rank) correlated with variables of our interest, i.e. K6, Q1, or Q2 scores.

We conducted structural equation modeling (SEM) [37] to estimate whether and how the session-by-session enthusiasm to complete homework could predict subsequent
session-by-session therapeutic outcomes. The session-by-session changes in K6 scores revealed, however, that most of the effect of the tCBT program was achieved by session #4: the change of mean K6 scores during the behavior therapy (sessions #2 through #4) was -1.75 points whereas that during the cognitive therapy (sessions #5 through #7 was -0.32 points only, which was less than one-fifth of that observed in the three sessions previously (Figure 1). Thus the participants with subthreshold depression had already responded well to the initial behavior therapy interventions and there was not much variability left throughout the latter half of the program (floor effect). We therefore chose to examine enthusiasm ratings as predictors of symptoms for the behavior therapy sessions.

Figures 2 and 3 represent two SEM models that we examined in this study to explain the relationships among these variables. In both models, the responses to the two enthusiasm scores were supposed to be determined using an endogenous variable termed “enthusiasm” to complete the homework for each session. The enthusiasm and the K6 score of the previous session were then used to predict the K6 score of the subsequent session. We also hypothesized that enthusiasm could be influenced by the K6 score of the same session. In Model 2, we further hypothesized that enthusiasm at a prior session was correlated with subsequent enthusiasm.

For both models, we examined two further submodels with or without constraints, whereby coefficients for the corresponding correlations were held constant across sessions (It was hypothesized that paths $a_2=a_3$, $b_2=b_3=b_4$, $c_2=c_3$, and $d_2=d_3$ in the constrained model, but not in the unconstrained model).

The goodness-of-fit of the models was evaluated using the following indexes: chi-square test of the model (considered to be not a bad fit if $P>0.05$), the Comparative Fit Index (CFI; considered to be a good fit if above 0.9), the Root Mean Square Error of Approximation Index (RMSEA; considered to be a good fit if below 0.05) and Tucker-Lewis Index (TLI; considered to be a good fit if above 0.95). We considered the
RMSEA as the main index because it has been shown to be more sensitive than the others [38].

There is no standard method to calculate sample sizes required for SEM. SEM usually requires large samples, but required sample sizes depend on various characteristics of the model as well as what differences one wishes to detect in the analyses. Thus some suggest that SEM models can perform well even with relatively small samples (e.g., 50 to 100) while a larger sample (e.g., n > 200) would be desirable [39].

All statistical tests were two-sided, and p values less than 0.05 were considered statistically significant. We used SPSS and AMOS (Version 21.0; IBM Inc., Armonk, NY, USA) for all analyses.

RESULTS

Patient characteristics

All in all 71 adults met the eligibility criteria for this study. Table 1 shows the demographic and clinical characteristics of the participants at baseline. The correlations between these baseline demographics (i.e., gender, age, job category, and job rank) and K6, Q1, Q2 scores at session 2 through 4 were in the ranges of 0.0 to 0.2 and were all non-significant.

Figure 1 showed change of the mean K6 score through the tCBT sessions. At baseline, the mean (range, SD) of the K6 score was 9.8 (5-18, 3.4), which dropped to 7.1 (0-22, 4.5) by session #4, with a pre-post effect size of 0.64. The mean (range, SD) of the first and the second questions to measure enthusiasm were 8.7 (4-10, 1.3) and 8.4 (5-10, 1.3) at session #2, 8.8 (6-10, 1.1) and 8.5 (5-10, 1.1) at session #3, and 8.6 (6-10, 1.3) and 8.1 (5-10, 1.3) at session #4, with Pearson correlation coefficients of 0.42, 0.46 and 0.57 between the first and second questions in the 2nd, 3rd and 4th session, respectively.
Five patients had dropped out by session #4. Five other patients had two missing values (Q1 and Q2 score for one of the sessions), and one patient had one missing value (Q2 score for session 4). Using the full information maximum likelihood method, we could include all participants in our analysis, and the final analysed sample size was 71.

Comparison of different models

Table 2 shows the goodness-of-fit indexes of the four competing models, as described in the Materials and Methods section. An acceptable fit was obtained for both the unconstrained and constrained versions of Model 2, with the constrained model showing a slightly better fit than the unconstrained model. Consequently, we chose this hypothesized model as best representing the session-by-session relationships between enthusiasm and depressive symptom.

Figure 4 shows the standardized coefficients for the final model. The correlation coefficients of the three paths were statistically significant, but path b was not ($P<0.001$ for path a, $P = 0.009$ for path c, and $P<0.001$ for path d, but $P = 0.96$ for path b). Thus, the K6 score measured at the beginning of the session did not influence the enthusiasm to complete the homework for that session. Both the K6 and the enthusiasm scores of the preceding session strongly and positively predicted the respective K6 and enthusiasm scores for the subsequent session, while enthusiasm to complete homework negatively predicted, i.e., decreased, the K6 score of the subsequent session.

DISCUSSION

The present study is the first to examine the session-by-session relationships between enthusiasm to engage in homework and changes in psychological distress through the course of behavior therapy for subthreshold depression using structural equation modeling. The best fitting model suggested that distress severity at the beginning of a
The session did not influence the enthusiasm for completing the homework from that session (estimated $r = 0.0$), which then, however, was negatively associated with the distress severity of the following session ($r = -0.21$ to $-0.24$).

The largest strength of this research is the examination of the session-by-session relationships between the enthusiasm to complete homework and psychological distress severity. A number of preceding studies have shown that homework in CBT was associated with the improvement of depression and anxiety; in these studies, however, homework compliance was measured either as average throughout the CBT sessions or at one out of the many sessions, while depression was typically measured at the end of treatment. In this study, by contrast, we examined the session-by-session relationships between homework and distress and found that enthusiasm to complete homework did indeed lead to a reduction in distress at the time of the subsequent session.

Another strength of our research is that we measured enthusiasm to engage in homework at the end of each session with regard to the particular homework assignments of that session. During the sessions, the therapists used Socratic questions to enhance the clients’ motivations while also modifying the assignments and troubleshooting possible barriers if necessary. In other words, in contrast with prior studies, which studied homework compliance in terms of the completion of the assigned homework *ex post facto*, we measured a variable that therapists can work on collaboratively with their clients and found that such enthusiasm did lead to distress symptoms reduction during the following week. It is also important to note that this enthusiasm was independent of the distress measured at the beginning of the session. In other words, therapists can strive to enhance the clients’ enthusiasm for homework regardless of the initial psychological distress severity.

It is important to note that not only distress, but also enthusiasm thus measured showed strong session-to-session correlations. It is natural to expect that distress severity at one session would predict distress severity at the subsequent session. The
very strong correlation between enthusiasm measurements throughout the sessions may be a reflection of the patient’s personality (e.g., tenacity to engage in assignments), his/her determination and preference for BT, and/or a stable collaborative relationship between the therapist and client. One could argue that there may be little room for the clinician to work on, if the former were the only factor behind the observed strong correlations. The observational nature of the current study precludes any further elucidation in this regard. However, our study does suggest that enhancing this factor through a collaborative therapist-patient relationship could contribute to a reduction in distress in the coming week.

There are several limitations in this study that should be acknowledged. First, as this research was a secondary analysis of a completed trial, the sample size was limited by the available data and was not as large as one would have preferred. This may have led to some possible type II errors (dismissing true associations). On the other hand, we were able to observe several statistically significant correlations that would be clinically meaningful. Second, as our sample consisted of employees with subthreshold depression who were not seeking clinical help and who were working, our results may not be readily generalizable to moderately or severely depressed clinical cases. Moreover, the majority of our samples were men, and this would also affect to generalizability. This limitation was primarily due to the high proportion of male employees in the company where the original RCT took place. Third, as the participants had made most of their improvement through the BT sessions alone, our session-by-session analyses of the relationships between homework enthusiasm and improvement in psychological distress were limited to the BT sessions and we could not investigate them through the CT sessions conducted in the latter half of the program. Thus our results would apply to BT homeworks but might not to CBT homeworks in general. Forth, although we estimated enthusiasm using two questions with satisfactory internal consistency reliability, as described in the Methods section, there is no
standardised method to measure enthusiasm, so more refinement in this direction is desirable. Fifth, client characteristics other than enthusiasm and level of distress, such as perfectionism and fear of failure [20], and therapist characteristics were not examined in the present study. Further research is needed to examine interplay of such characteristics for homework completion and improvement of psychological distress or depression.

In conclusion, the clinical and research implications of the present study may be as follows. Clinicians now have empirical evidence when they strive to work collaboratively with their clients to enhance their enthusiasm for homework, since it was shown that a higher enthusiasm leads to a reduction in distress during the subsequent week; moreover, they can and should do this regardless of the clients’ distress severity at that session because these two factors are not correlated. Research into homework in CBT should no longer be content in making global measurements once at the end of the treatment, but such research should examine its vicissitudes and mutual influences throughout the sessions using statistical methods that allow longitudinal modeling. Only then can such research illuminate what each clinician and patient can do in each session.

Abbreviations:
tCBT; Cognitive behavioral therapy via telephone
SEM; Structural equation modeling
EAP; Employee assistance program
BDI2; Beck Depression Inventory-II
CIDI; Composite International Diagnostic Instrument
CFI; Comparative Fit Index
RMSEA; Root Mean Square Error of Approximation Index
Competing interests:

HY, IH and HM have no competing interests to declare.

FTA has received lecture fees from Eli Lilly, Meiji, Mochida, MSD, Pfizer and Tanabe-Mitsubishi, and consultancy fees from Sekisui and Takeda Science Foundation. He is diplomate of the Academy of Cognitive Therapy. He has received royalties from Igaku-Shoin, Seiwa-Shoten and Nihon Bunka Kagaku-sha. The Japanese Ministry of Education, Science, and Technology, the Japanese Ministry of Health, Labor and Welfare, and the Japan Foundation for Neuroscience and Mental Health have funded his research projects.


Authors’ contributions:
HY and FTA conceived and designed the experiments. FTA, KN and HM contributed to acquisition of the data. HY managed the data. HY, FTA and ST analyzed the data. All authors commented on drafts of the manuscript. All authors read and approved the final manuscript.

Authors’ information:
HY and IH are psychiatrist and PhD student at Kyoto University Graduate School of Medicine.
FTA is psychiatrist and professor at Department of Health Promotion and Human Behavior, Kyoto University Graduate School of Medicine.
ST is statistician and associate professor at Department of Management Science, Faculty of Engineering, Tokyo University of Science.
KN is psychiatrist and professor at Department of Mental Health, Tokyo University Graduate School of Medicine.
HM is psychologist and specialist in CBT.

Acknowledgments:
GENKI Project collaborators include:
Principal Investigator:
– Toshiaki A. Furukawa, MD, PhD, Kyoto University Graduate School of Medicine/School of Public Health, Department of Health Promotion and Human Behavior, Kyoto, Japan
Co-principal Investigators:
– Masaru Horikoshi, PhD, National Center of Neurology and Psychiatry, Kodaira, Japan
– Norito Kawakami, MD, PhD, Department of Mental Health, University of Tokyo Graduate School of Medicine, Tokyo, Japan

Project and Data Managers:
– Masayo Kadota, RN, MSc, Department of Psychiatry and Cognitive-Behavioral Medicine, Nagoya City University Graduate School of Medical Sciences, Nagoya, Japan
– Megumi Sasaki, PhD, Center for Education and Research on the Science of Preventive Education, Naruto University of Education, Naruto, Japan

Investigators:
– Yuki Sekiya, MA, University of Tsukuba Graduate School of Comprehensive Human Science, Tsukuba, Japan.
– Hiroki Hosogoshi, PhD, Faculty of Clinical Psychology, Kyoto Bunkyo University, Kyoto, Japan
– Masami Kashimura, PhD, Institute of Psychology, University of Tsukuba Graduate School of Comprehensive Human Sciences, Tsukuba, Japan
– Kenichi Asano, PhD, Faculty of Applied Psychology, Tokyo Seitoku University, Tokyo, Japan
– Hitomi Terashima, PhD, Institute of Psychology, University of Tsukuba Graduate School of Comprehensive Human Sciences, Tsukuba, Japan
– Kazunori Iwasa, PhD, Faculty of Education, Shujitsu University, Okayama, Japan
– Minoru Nagasaku, PhD, Faculty of Psychology, Surugadai University, Hanno, Japan
– Takaki Fukumori, PhD, Division of Psychology and Health, University of Tokushima Institute of Socio-Arts and Sciences, Tokushima, Japan
– Madoka Niwa, MA, Institute of Women’s Health, Tokyo Women’s Medical University, Tokyo, Japan
– Yuki Oe, MA, University of Tsukuba Graduate School of Comprehensive Human Sciences, Tsukuba, Japan
– Maki Shibata, BA, Kitabayashi Hospital, Nagoya, Japan
– Hiroko Fujisato, PhD, Faculty of Human Sciences, Mejiro University, Tokyo, Japan
– Junko Ito, RN, Hoshigaoka Maternity Clinic, Nagoya, Japan
– Chiori Hirota, MA, Tsuchiura Kyodo General Hospital, Tsuchiura, Japan
– Naoki Kawasaki, PhD, School of Human Services, Hokusho University, Ebetsu, Japan
– Issei Shinmei, MA, National Center of Neurology and Psychiatry, Kodaira, Japan
– Yuriko Takagishi, MA, Surugadai University Counseling Center, Hanno, Japan
– Shihori Yamada, CCW, Nagoyaka House Dekimachi, Nagoya, Japan
– Masato Hattori, MA, Tokyo Juvenile Classification Home, Tokyo, Japan
– Saori Kitagawa, Sapporo, Japan
– Kanako Nakazawa, MA, National Center of Neurology and Psychiatry, Kodaira, Japan
– Kyoko Shimada, MPH, Department of Mental Health, University of Tokyo Graduate School of Medicine, Tokyo, Japan
– Haruki Shimoda, BSc, Department of Mental Health, University of Tokyo Graduate School of Medicine, Tokyo, Japan
– Masao Tsuchiya, PhD, National Institute of Occupational Safety and Health, Kawasaki, Japan
– Maki Umeda, RN, MA, MPH, Department of Mental Health, University of Tokyo Graduate School of Medicine, Tokyo, Japan
– Ayano Yamagishi, MA, University of Tsukuba Graduate School of Comprehensive Human Sciences, Tsukuba, Japan

Statistician:
– Louis C. Grothaus, MA, Group Health Research Institute, Group Health, Seattle, USA
References


10. Barth J, Munder T, Gerger H, Nuesch E, Trelle S, Znoj H, Juni P,


38. Hu L-t, Bentler PM: *Fit indices in covariance structure modeling: Sensitivity to underparameterized model misspecification.*
Table 1. Demographic and clinical characteristics of the participants.

<table>
<thead>
<tr>
<th>Gender</th>
<th>Male</th>
<th>58 (81.7%)</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>Female</td>
<td>13 (18.3%)</td>
</tr>
<tr>
<td>Age</td>
<td>Mean (Range, SD)</td>
<td>39.0 (23-57, 8.0)</td>
</tr>
<tr>
<td>Job category</td>
<td>Sales/Marketing</td>
<td>25 (35.2%)</td>
</tr>
<tr>
<td></td>
<td>Production/Factory</td>
<td>12 (16.9%)</td>
</tr>
<tr>
<td></td>
<td>Engineering/Technical</td>
<td>18 (25.4%)</td>
</tr>
<tr>
<td></td>
<td>Administration/Management</td>
<td>15 (21.1%)</td>
</tr>
<tr>
<td></td>
<td>Unknown</td>
<td>1 (1.4%)</td>
</tr>
<tr>
<td>Job rank</td>
<td>Supervisory</td>
<td>18 (25.4%)</td>
</tr>
<tr>
<td></td>
<td>Nonsupervisory</td>
<td>53 (74.6%)</td>
</tr>
<tr>
<td>K6 score</td>
<td>Mean at baseline (Range, SD)</td>
<td>9.8 (5-18, 3.4)</td>
</tr>
<tr>
<td>Enthusiasm</td>
<td>Mean of Q1 at baseline (Range, SD)</td>
<td>7.0 (1-10, 2.2)</td>
</tr>
<tr>
<td></td>
<td>Mean of Q2 at baseline (Range, SD)</td>
<td>7.6 (0-10, 2.0)</td>
</tr>
</tbody>
</table>
Table 2. Goodness-of-fit indexes for the four competing models

<table>
<thead>
<tr>
<th></th>
<th>P value for Chi-squared test</th>
<th>CFIc</th>
<th>RMSEAd</th>
<th>TLIe</th>
</tr>
</thead>
<tbody>
<tr>
<td>Model 1, unconstrained</td>
<td>&lt;0.001</td>
<td>0.601</td>
<td>0.197</td>
<td>0.219</td>
</tr>
<tr>
<td>Model 1, constraineda</td>
<td>&lt;0.001</td>
<td>0.608</td>
<td>0.180</td>
<td>0.346</td>
</tr>
<tr>
<td>Model 2, unconstrained</td>
<td>0.366</td>
<td>0.990</td>
<td>0.033</td>
<td>0.978</td>
</tr>
<tr>
<td>Model 2, constrainedb</td>
<td>0.400</td>
<td>0.992</td>
<td>0.025</td>
<td>0.987</td>
</tr>
</tbody>
</table>

a Hypothesized paths a, b and c were constant throughout the sessions
b Hypothesized paths a, b, c, and d were constant throughout the sessions
c Comparative Fit Index; considered to be a good fit if above 0.9
d Root Mean Square Error of Approximation Index; considered to be a good fit if below 0.05
e Tucker-Lewis Index; considered to be a good fit if above 0.95